Author's response to reviews

Title: Irradiation-induced telomerase activity and gastric cancer risk: a case-control analysis in a Chinese Han population

Authors:

Xianli He (wanghe@fmmu.edu.cn)
Qing Qiao (qqing@gmail.com)
Naijian Ge (gelarge@163.com)
Jing Nan (nanjing0507@163.com)
Shuqun Shen (shenshuqunssq@sohu.com)
Zizhong Zhang (xskxue@fmmu.edu.cn)
Yefa Yang (TISC@EHBH.cn)
Guoqiang Bao (guoqiang@fmmu.edu.cn)

Version: 4 Date: 7 June 2010

Author's response to reviews:

Dear editors,

This is the finally revised version of our manuscript. All the demanded changes were re-writted in the revised manuscript.

Thanks,
Guoqiang Bao